Author's response to reviews

Title: Benign Cystic Mesothelioma of the Appendix

Authors:

Donal B O' Connor (donaloconor@yahoo.com)
David Beddy (davbeddy@rcsi.ie)
Muyiwa A Aremu (m.aremu@rcsi.ie)

Version: 4 Date: 28 September 2010

Author's response to reviews: see over
Dear Editor,

Thank you for accepting our revised manuscript for publication in your journal. We have revised the manuscript according to your requested revision.

Editor’s comment:

“Include in the abstract as well as in the discussion that this is the first case of a benign cystic mesothelioma arising from the appendix and complicating a presentation of acute appendicitis.”

This has now been stated in both abstract and conclusion.

Yours sincerely,

Donal B O’ Connor,

Surgical Research Registrar
St Vincent’s University Hospital